Author's response to reviews

Title: Acceptability of intrapartum HIV counselling and testing in Cameroon

Authors:

Eugene J Kongnyuy (kongnyuy@gmail.com)
Enow R Mbu (rembu2000@yahoo.com)
F X Mbopi-Keou (fxmkeou@hotmail.com)
Nelson N Fomulu (fomulunelson@yahoo.fr)
Philip N Nana (njotang@yahoo.fr)
Pierre M Tebeu (pmtebeu@yahoo.fr)
Rebecca N Tonye (rtonye@yahoo.com)
Robert JI Leke (robert.leke@camnet.cm)

Version: 6 Date: 31 January 2009

Author's response to reviews:

Dear Editor,

Thanks for giving us the opportunity to resubmit the manuscript. We have addressed all the comments raised by reviewers in the current version of the manuscript.

The abstract has been revised and additional information added. In particular more context has been added in the background and the methods section enhanced by including the details of the counseling and methods of analyses.

Yours sincerely,

Eugene Kongnyuy